Author's response to reviews

Title: Influenza vaccination of primary healthcare physicians may be associated with vaccination of their patients: a vaccination coverage study

Authors:

Pere Godoy (pere.godoy@gencat.cat)
Jesus Castilla (jesus.castilla.catalan@cfnavarra.es)
Jose M Mayoral (josem.mayoral.sspa@iuntadeandalucia.es)
Vicente Martin (vicente.martin@unileon.es)
Jenaro Astray (jenaro.astray@salud.madrid.org)
Nuria Torner (nuria.torner@gencat.cat)
Diana Toledo (pmgripea@ub.edu)
Nuria Soldevila (nuriasolde_@hotmail.com)
Fernando Gonzalez-Candelas (fernando.gonzalez@uv.es)
Susana Garcia-Gutierrez (SUSANA.GARCIGUTIERREZ@osakidetza.net)
Jose Diaz-Borrego (jose.diazborrego.sspa@juntadeandalucia.es)
Sonia Tamames (TamGomSo@jcyI.es)
Angela Dominguez (angela.dominguez@ub.edu)

Version: 6 Date: 7 March 2015

Author's response to reviews:

Dear Mr. Gerard Clunn:

We are pleased to enclose the revised manuscrit entitled “Influenza vaccination of primary healthcare physicians may be associated with vaccination of their patients: a vaccination coverage study” with the modifications suggested by the editor. We believe that all the modifications suggested have been incorporated. We thank the editor and the reviewers for their helpful suggestions which we believe have substantially improved the article.

All points are answered in detail below.

Yours sincerely

Dr Pere Godoy

MS: 3345389581505314

Influenza vaccination of primary healthcare physicians may be associated with vaccination of their patients: a vaccination coverage study

Pere Godoy, Jesus Castilla, Jose M Mayoral, Vicente Martin, Jenaro Astray, Nuria Torner, Diana Toledo, Nuria Soldevila, Fernando Gonzalez-Candelas, Susana Garcia-Gutierrez, Jose Diaz-Borrego, Sonia Tamames and Angela Dominguez

Dear Dr Godoy,
Thank you for your submission to BMC Family Practice. Peer review of your manuscript is now complete.

On the basis of the comments we have received we feel that your manuscript is suitable for publication in the journal. However, before we proceed with acceptance there are some further editorial revisions which we require you to make. Please note that your manuscript cannot be accepted for publication until these revisions have been satisfactorily addressed.

Further editorial revisions:
- Format abs sections according to our guidelines.
  We have formatted the abstract to guidelines of BMC Family Practice
- Delete Funding section and add funding info to the Acknowledgement section
  We have done it
We would be grateful if you could address the comments in a revised manuscript and provide a cover letter giving a point-by-point response to the concerns.

Please also ensure that your revised manuscript conforms to the journal style (http://www.biomedcentral.com/info/ifora/medicine_journals ). It is important that your files are correctly formatted.

We have done it